Great Basin Incident Commander Conference Call Agenda 2016

Date: Tuesday, August 02,2016 | Note taker: Gina Bald

GB MAC/GACC Coordinator:

Agenda/Incident Commander Discussion Points: Summarize the following in 4 min or less:

. Fire Size / % contained / # of people / Cost to date

. Significant changes since the last 209

. BRIEF fire synopsis /Critical Incident Information (use Directional vs DIVS’s)
. Political Issues / Community Meetings

o Closures / Evacuations / Public Safety Issues

. Major Safety Concerns / Injuries / Accidents

. Critical Resource Needs

. Resources Available to Reassign/Share/Demob

Predictive Services:
Weather update:

Cold front in GB from N-S next 18 hrs. Wind shift late afternoon over Pioneer
and VA Mtn complex. Strong winds in ID. Frontal passage early morning on

Gina McGuire

Pioneer. Won't die down tonight. Moving into N. UT tomorrow morning. VA
Mtn Complex will have winds N-NE after midnight moving NE. Precipitation
in UT, E. half of NV. Already lightning in NE NV. Dryin ID. Some new
starts. Next couple of days lightning suppressed. Thursday — return of
lightning.

GBCC Update: e 21 news starts for 288

Gina Dingman e large Fire Growth: 23327 acres

e GB moved into #1 GACC on IMSR

e 1VLAT, 2 CL-214, 3 ATs, 22 SEATs (1 D/0), 2 Bravos, 1 LPs
e Committed in GB; 121 crews (19 T1), 216 Engines

o No resources to give up.

e Reassigning resources as quickly as possible

1 52119 /40% /791 /2.8M
Virginia Mts Complex e Changes since 209 submitted: Restart of evacuations
IC: Tim Roide e 7 Lakes: Contained. Suppression repair

e Sage: Contained. Suppression repair

e Tule: Very active. High resistance. Strong air support. Structures
prepped. Evacuation Notices distributed. North Flank
progression. Woods Ranch — structure protection. No issues. Big
Canyon — evacuation notices. Looking art alternate strategies
other than direct.

e Anderson: Very active. North and south activity. Structures

prepped.
e Political issues: Moderate interest




Evacuations: Big Canyon and Woods Ranch (Tule)

Pyramid Lake closed to watercraft

No power in Sutcliff

No injuries

Critical Resource Needs: None

Resources available: Incremental over next several days —
Engines and T2 Crews

2

Fire: Pioneer

IC: Beth Lund
Report: Ron Bollier

42031/ 28% /1423 /14.AM

Changes since 209 submitted: 4000 acres

West: Mop-up and patrol

North: Rock Creek still an issue. 3.8 miles from Lowman. MAPS
in place for Lowman. Structure protection group in place with
VFDs. DIV F firing operations to tie off Rock Creek.

Southeast: Hit Rabbit Creek fire. Successful burnout tying Pilot
Peak and Sunset Peak together to Hwy 21.

Critical Resource Needs: 3 T1 Crews

Closure still good. Restrictions in Hwy 17 corridor.

No evacuations

Injuries: 7 to date, 1 loss time

Public meeting on Lowman

Resources available for release: None

3

Poodle

IC: J. Henry
Report: Glen Uhlig

5200/ 60% /410 /1M

Very rocky. Tools and Dozers broken.

Critical Resource Needs: None

No injuries

Red Flag Warning today

Resources Available: None today, possibly tomorrow

4
Fire: Cliff Creek
IC: Tony DeMasters
IC2-T Evans Kuo

29429/ 84% / 692 /11.5M

Changes since last 209 submitted: 411 acre growth. Team
Transition from Demasters to Bird

West: Looking good. Planning for re-entry.

South: Patrol by air

North: Natural barriers

East: Completed burning operations

Wind event today. Will look at interior islands.

Had Town meeting in Bonderant: Discussed re-entry and air quality
Political Issues: None

Unified Command dissolved

Closures: No changes

Injury/Accident: None

Critical Resource Needs: None

Resources to reassign: None

5
Fire: Comet

358/

82% / 45 / $824,000
Looking good. Winding down




IC: McPhetridge
Report: Terry
Swinscoe

e No critical resource needs
e Noinjuries

6

Fire: Line

IC: Ahivers
Report: No Report

832 /80% / 144 / $275,000

7

Fire: Hospital Bar
IC: Rasmussen
Report: No Report

102 / 41% /20 / $10,000
e Full containment tonight

8
Fire: Box Canyon

1485 /0% /76 /221,000
e Changes from last 209 submitted: N. UT T3 ordered. Will shadow

IC: Knudson and take over Wednesday.
Report by Brook e Not active. Hells Kitchen growth.
Chadwick e Possible PSD work.
e WFM today for situational awareness
e Cooperator today at 1000 with local officials
e Public meeting still possible
e Closures: Enlarged closure area
e Noinjuries
e Critical Resource Needs: T2 Crew to replace one reassigned to
new start Sheep Creek
e Resources to assign: None
MAC / GBCC This is the first day of crew strategy. Two hotshots distributed.

Coordinator
Derrek Hartman

Date/Time of Next Call:

8/3 @ 0730 MT




